Author’s response to reviews

Title: Trochanteric osteotomy versus posterolateral approach: function the first year post surgery. A pilot study

Authors:

- Margot van der Grinten (m.vandergrinten@erasusmc.nl)
- Max Reijman (m.reijman@erasusmc.nl)
- Frans C. van Biezen (f.vanbiezen@erasusmc.nl)
- Jan A.N. Verhaar (j.verhaar@erasusmc.nl)

Version: 10 Date: 23 June 2011

Author’s response to reviews:

Dear editor,

in this final version changes have been made according to your suggestions:

the title in the submission system is identical to the title of the manuscript
the list of authors has been rewritten according to your suggestions and they are all linked by the number 1.
The tables and the figure were already changed in the previous version.
the authors contribution has been changed into a paragraph.

Yours sincerely,
Margot van der Grinten